Author's response to reviews

Title: Reduced thoracolumbar fascia shear strain in human chronic low back pain

Authors:

Helene M Langevin (helene.langevin@uvm.edu)
James R Fox (james.fox@uvm.edu)
Cathryn Koptiuch (cathryn.koptiuch@uvm.edu)
Gary J Badger (bagy.badger@uvm.edu)
Ann C Greenan-Naumann (Ann.Greenan-Naumann@vtmednet.org)
Nicole A Bouffard (nicole.bouffard@uvm.edu)
Elisa E Konofagou (ek2191@columbia.edu)
Wei-Ning Lee (w12134@gmail.com)
John J Triano (jtriano@cmcc.ca)
Sharon M Henry (sharon.henry@uvm.edu)

Version: 7 Date: 2 September 2011

Author's response to reviews:

Formatting changes were made as requested